Author's response to reviews

Title: Dopamine transporter 3'UTR VNTR genotype is a marker of performance on executive function tasks in children with ADHD

Authors:

Sherif Karama (sherif.karama@mail.mcgill.ca)
Natalie Grizenko (Natalie.grizenko@douglas.mcgill.ca)
Edmund Sonuga-Barke (ejb3@soton.ac.uk)
Alysa Doyle (doylea@helix.mgh.harvard.edu)
Joseph Biederman (biederman@helix.mgh.harvard.edu)
Valentin Mbekou (valentin.mbekou@douglas.mcgill.ca)
Anna Polotskaia (anna.polotskaia@douglas.mcgill.ca)
Marina Ter-Stepanian (marina.terstepanian@douglas.mcgill.ca)
Rosherrie De Guzman (rosherrie.deguzman@douglas.mcgill.ca)
Johanne Bellingham (beljoh@douglas.mcgill.ca)
Sarojini Sengupta (sarojini.sengupta@douglas.mcgill.ca)
Ridha Joober (ridha.joober@douglas.mcgill.ca)

Version: 3 Date: 5 June 2008

Author's response to reviews: see over
Dear Editors,

Here are the changes to the manuscript.

All requested formatting changes have been made. Please note that the figures were for review purposes only and were not meant to be published. This was explicit in the response to reviewers; the tables were simply to show to the reviewer that results were not due to outliers.

Further few minor corrections/modifications have also been implemented.

Abstract

p.2, TOL, FFDI, SOPT, and WCST abbreviations have been removed and replaced by the full name of the tests.

Introduction

p.4, First paragraph,

Last phrase, DAT10 was replaced by DAT1 10

p.4 Second paragraph,

“This being said” was added prior to “A few recent…”

“Restructured phrase beginning with “For instance, Cornish…”

“Further” was added to phrase beginning with “A recent study…”

Reference 24 was moved up

p.5 First paragraph,

"that” (second word) was replaced by “than”
Comma was placed after 10/10 genotype

p.6 Second paragraph,

“though variable,” was removed

p.6 Third paragraph

VNTR and UTR were defined

p.7 Second paragraph

“Interview” was replaced by “Intelligence”

p.10 Second paragraph

“of the WISC-III” was removed

Discussion

p.13 Third paragraph,

4th line, “children with the” was replaced by “carriers of the”

6th line, “findings” was replaced by “these”

p.15 Third paragraph

“DAT1 genotype” was replaced by “DAT1 gene genotype”

Tables

Number of decimals after the last 0 was adjusted.

Best Regards,
Sherif Karama (for the authors)
McGill University,
Department of Psychiatry,
Research and Training building,
1033 Pine Avenue West,
Montreal, Quebec,
H3A 1A1